Author's response to reviews

Title: Functional microRNA high throughput screening reveals miR-9 as a central regulator of liver oncogenesis by affecting the PPARA-CDH1 pathway

Authors:

Alexandra Drakaki (adrakaki@mednet.ucla.edu)
Maria Hatziapostolou (MHatziapostolou@mednet.ucla.edu)
Christos Polytarchou (CPolytarchou@mednet.ucla.edu)
Christina Vorvis (cvorvis@ucla.edu)
George A Poultsides (gpoultsides@stanford.edu)
John Souglakos (johnsougl@gmail.com)
Vassilis Georgoulias (georgouliasv@gmail.com)
Dimitrios Iliopoulos (diliopoulos@mednet.ucla.edu)

Version: 2
Date: 10 February 2015

Author's response to reviews:

Dear Ms. Battad,

thank you for your email and comments. We have made all the appropriate corrections in the manuscript according to your suggestions. Specifically, we have include line and page numbers throughout the manuscript, included a statement in the methods about ethical approval of our study, included all authors' emails and added competing interest, authors contributions and acknowledgement sections.

Sincerely,

Dimitrios Iliopoulos